Author’s response to reviews

Title: A combined community- and facility-based approach to improve pregnancy outcomes in low-resource settings: a Global Network cluster randomized trial

Authors:

Omrana Pasha (omrana.pasha@aku.edu)
Elizabeth M McClure (mcclure@rti.org)
Linda L Wright (wrightl@mail.nih.gov)
Sarah Saleem (sarah.saleem@aku.edu)
Shivaprasad S Goudar (sgoudar@jnm.edu)
Elwyn Chomba (echomba@zamnet.zam)
Archana Patel (dr.patel@yahoo.com)
Fabian Esamai (fesamai2007@gmail.com)
Ana Garces (ana.garces@imsalud.org)
Fernando Althabe (althabef@gmail.com)
Bhala Kodkany (dr.kodkany@jnm.edu)
Hillary Mabeya (hmabeya@yahoo.com)
Albert Manasyan (Albert.Manasyan@cidrz.org)
Waldemar A Carlo (wcarlo@peds.uab.edu)
Richard J Derman (rderman@christianacare.org)
Patricia L Hibberd (phibberd@partners.org)
Edward K Liechtly (eliecht@iupui.edu)
Nancy Krebs (Nancy.krebs@ucdenver.edu)
K M Hambidge (Michael.Hambidge@ucdenver.edu)
Pierre Buekens (pibuekens@tulane.edu)
Janet Moore (jhartz@rti.org)
Alan H Jobe (Alan.Jobe@cchmc.org)
Marion Koso-Thomas (kosomari@mail.nih.gov)
Dennis D Wallace (dwallace@rti.org)
Suzanne Stalls (sstalls@acnm.org)
Robert L Goldenberg (rlg88@columbia.edu)

Version: 3 Date: 16 August 2013

Author’s response to reviews:

August 10, 2013

Dear Editor,

We are pleased to submit the revised manuscript, “A combined community- and facility-based approach to improve pregnancy outcomes in low-resource settings: a Global Network cluster randomized trial” which we hope addresses the reviewer comments, as included in our point-by-point response. In response to the reviewer 2, we added a number of paragraphs further documenting how the intervention was carried out and the oversight provided by the study team.
We hope that the revision is sufficient, but would be pleased to work with the editor if further changes are felt to be necessary prior to publication.

Sincerely,

Omrana Pasha, MD, Elizabeth McClure, PhD, and Robert Goldenberg MD, on behalf of the EMONC Trial investigators